CONTRIBUTI DI RICERCA CRENOS



## GREEN LICENSES AND ENVIRONMENTAL CORRUPTION: A RANDOM MATCHING MODEL

Angelo Antoci Simone Borghesi Gianluca lannucci

## **WORKING PAPERS**

2016/15

#### CENTRO RICERCHE ECONOMICHE NORD SUD (CRENOS) Università di Cagliari Università di Sassari

CRENOS was set up in 1993 with the purpose of organising the joint research effort of economists from the two Sardinian universities (Cagliari and Sassari) investigating dualism at the international and regional level. CRENoS' primary aim is to improve knowledge on the economic gap between areas and to provide useful information for policy intervention. Particular attention is paid to the role of institutions, technological progress and diffusion of innovation in the process of convergence or divergence between economic areas. To carry out its research, CRENoS collaborates with research centres and universities at both national and international level. The centre is also active in the field of scientific dissemination, organizing conferences and workshops along with other activities such as seminars and summer schools.

CRENoS creates and manages several databases of various socio-economic variables on Italy and Sardinia. At the local level, CRENoS promotes and participates to projects impacting on the most relevant issues in the Sardinian economy, such as tourism, environment, transports and macroeconomic forecasts.

www.crenos.it info@crenos.it

> CRENOS - CAGLIARI VIA SAN GIORGIO 12, I-09100 CAGLIARI, ITALIA TEL. +39-070-6756406; FAX +39-070- 6756402

> > CRENOS - SASSARI VIA MURONI 23, I-07100 SASSARI, ITALIA TEL. +39-079-213511

Title: GREEN LICENSES AND ENVIRONMENTAL CORRUPTION: A RANDOM MATCHING MODEL

First Edition: November 2016

# Green licenses and environmental corruption: a random matching model

Angelo Antoci<sup>a</sup>, Simone Borghesi<sup>b</sup>, Gianluca Iannucci<sup>c</sup>

<sup>a</sup>Department of Economics and Business, University of Sassari, Italy <sup>b</sup>Department of Political and International Sciences, University of Siena, Italy <sup>c</sup>Department of Economics and Management, University of Florence, Italy

November 15, 2016

#### Abstract

This paper studies environmental corruption via a random matching evolutionary game between a population of firms and a population of bureaucrats who have to decide whether to release a "green" license to the firms. A firm obtains the license if the bureaucrat checks that it complies with environmental regulations, otherwise it is sanctioned. The model assumes that there are two types of bureaucrats (honest and dishonest), two types of firms (compliant and non-compliant), and two possible crimes (corruption and extortion). Corruption occurs when a dishonest bureaucrat accepts a bribe from a non-compliant firm, while extortion occurs when a dishonest bureaucrat claims a bribe from a compliant firm. When there is no dominance of strategies, we show that there exist two bistable regimes, in which two attractive stationary states exist, and two regimes with an internal stable equilibrium, corresponding to the mixed strategy Nash equilibrium of the one-shot static game, surrounded by closed trajectories. From comparative statics analysis performed on the latter two dynamic regimes, it emerges that policy instruments may help the Public Administration reduce both corruption and extortion, although increasing sanctions and detection probability do not always get the desired results.

Keywords: Bureaucratic corruption, Evolutionary games, Environmental regulations, Economics of crime.

JEL classification: C73, D21, D73, K42, Q52.

## 1 Introduction

Recently, the media have focused a lot of attention on compliance with environmental regulations by industrial enterprises and on inspection by bureaucrats, especially after the so called *Volkswagen scandal.*<sup>1</sup> However, these issues have attracted the attention of economists for a

*E-mail addresses*: angelo.antoci@virgilio.it (Angelo Antoci), simone.borghesi@unisi.it (Simone Borghesi), gianluca.iannucci@unifi.it (Gianluca Iannucci).

<sup>&</sup>lt;sup>1</sup>In September 2015, the German automaker Volkswagen Group has received a notice of violation of the Clean Air Act from the United States Environmental Protection Agency (EPA). It was found that the enterprise had used a software applied to some diesel engines to activate certain emissions controls only during laboratory emissions testing. The software caused the compliance with U.S. environmental standards for nitrogen oxide  $(NO_x)$  output during laboratory tests, but produce up to 40 times higher  $NO_x$  output in real-world driving. See, for further details, both links: Financial Times and The New York Times.

long time. The incentive to comply with environmental regulation is obviously reduced if the inspection effort is insufficient or not properly done. This may depend on many reasons, including lack of financial resources to perform proper controls, inability of inspectors to discover violations or corruption of inspectors. In particular, several studies have analysed the negative effects of corruption on environmental degradation. In this regard, it is possible to identify two main strands in the literature on this issue. The first strand has examined the effects of bureaucracy and lobbying groups on the efficacy of environmental policy (Lopez and Mitra, 2000; Damania et al., 2003; Fredriksson et al., 2003; Fredriksson and Svensson, 2003; Cole et al., 2006; D'Amato and Zoli, 2012, 2013; D'Amato et al., 2015). The second strand, instead, has analyzed the effects of corruption on environmental degradation through economic growth, investigating how corruption affects the shape of the so-called *Environmental Kuznets Curve* (Welsch, 2004; Cole, 2007; Leitao, 2010).<sup>2</sup>

The link between corruption and economic growth is the object of a heated debate in the literature. Several studies (Mauro, 1998; Mo, 2001; Blackburn et al., 2006) consider bribery as a "grabbing hand" and, therefore, as an evil for economic growth. This seems to be confirmed by the fact that the most corrupt countries have generally low income level (Svensson, 2005), which is usually explained by the literature via the negative role that bad institutions can play for economic growth (Lipset, 1960; Demsetz, 1967; Treisman, 2000; Glaeser et al., 2004; Acemoglu et al., 2012).<sup>3</sup> Some authors, however, propose a different viewpoint that conceives corruption as a "helping hand" and argue that bribes may lead firms to allocate their resources more efficiently, in an economy afflicted by slow bureaucracy and rigid laws (see, e.g., Leff, 1964; Huntington, 1968; Lui, 1985). Finally, some contributions take a more intermediate position emphasizing that corruption has non-linear effects on economic growth that depend on the quality of the institutions (Méndez and Sepúlveda, 2006; Aidt, 2009; Méon and Weill, 2010).

Corruption has attracted much attention also in game theory that generally models this phenomenon as the result of a strategic interaction between at least two kinds of players.<sup>4</sup> Among game theoretic models on corruption, a particularly interesting contribution is provided by the so-called *inspection games*,<sup>5</sup> where one player, a policeman, must decide whether to inspect the other player, who in turn must decide whether to infringe a regulation (Tsebelis, 1989, 1990). According to Holler (1993), the inspection games have no Nash equilibrium in pure strategies since both players have the possibility to improve their payoff values by choosing an alternative strategy, given the strategy of the other player. Therefore, there is a single mixed strategy

<sup>&</sup>lt;sup>2</sup>As it is well-known, the Environmental Kuznets Curve (EKC) is an inverted U-shape relationship between environmental degradation and per capita income, which suggests that environmental degradation first increases at early stages of economic growth and then decreases when income exceeds a certain level. The name was coined due to its similarity to the work of Kuznets (1955) on the relationship between income inequality and per capita income (cf. Grossman and Krueger, 1991; Shafik and Bandyopadhyay, 1992; Panayotou, 1993). See Borghesi (2001), Dinda (2004) and Kijima et al. (2010) for extensive reviews and in-depth discussions on the EKC.

<sup>&</sup>lt;sup>3</sup>More precisely, corruption is generally found to have detrimental effects on economic growth, both directly and indirectly through lower levels of investments, openness, schooling and political stability (Pellegrini and Gerlagh, 2004). See Dreher and Herzfeld (2008) for a survey on the economic costs of corruption and a discussion of its transmission channels.

 $<sup>^{4}</sup>$ An alternative (non-game theoretic) approach that is sometimes used to study crime deterrence in general is *decision theory* that involves only one actor (see, for further details, Becker, 1968; Garoupa, 1997; Polinsky and Shavell, 2000). We will not adopt this approach here since we focus more specifically on corruption that is an agreement between at least two actors: a player who decides to infringe a regulation and, to avoid being sanctioned, offers a bribe to another player who accepts the bribe and decides not to sanction the former player.

<sup>&</sup>lt;sup>5</sup>Another way to model corruption adopting a game theoretic framework is the *principal-agent theory*, where crime occurs due to the presence of asymmetric information between the principal, usually the Public Administration, and the agent, a public official (see, for further details, Bardhan, 1997; Acemoglu and Verdier, 2000; Di Gioacchino and Franzini, 2008). We do not use this setting here since the focus of the present analysis is not on the existence of asymmetric information across agents but on the mechanism underlying the choices of randomly matched agents in an evolutionary context.

Nash equilibrium that has counter-intuitive comparative statics properties (Andreozzi, 2004). In fact, these models show that increasing sanctions and the probability of being discovered for the player who must decide whether to infringe a norm, have no effects on law enforcement.<sup>6</sup> The only way to reduce crimes is to increase the inspection incentives for the policemen. Andreozzi (2004) proposes a variant of the 2x2 simultaneous-move inspection game originally set forth by Tsebelis (1989), namely, a sequential version in which the inspector acts as a Stackelberg leader and shows that in that case the opposite result applies: in order to reduce crimes policymakers should decrease (rather than increase) the inspector's incentive to catch the transgressor. However, as Friehe (2008) has shown, if the enforcer and the offender payoffs are correlated, increasing the catch bonus cannot increase crime while the latter can be decreased by increasing the severity of the sanction, thus contradicting the counterintuitive results that emerged from both the simultaneous and sequential versions of the game described above.

In this paper we adopt the framework of the inspection games using an evolutionary context (see, e.g., Andreozzi, 2002). The *evolutionary game theory* assumes that large populations of players with bounded rationality learn, imitate, and adopt the relatively more rewarding strategies. As pointed out by Cressman et al. (1998), this context seems particularly appealing for the study of crime, since it captures the positive influence of good role models in society that is often stressed as an important factor for reducing crime.

Differently from other evolutionary models on corruption that study the dynamics of only one population, either of firms playing a public contracting game or of public officials (cf., Antoci and Sacco, 1995, 2002, respectively), we propose a random-matching evolutionary game between a population of firms and a population of bureaucrats.<sup>7</sup> In each instant of time, there is a large number of random pairwise encounters between firms and bureaucrats. In each encounter a bureaucrat checks the compliance with environmental regulations by a firm. When the environmental laws are respected, the firm obtains a "green" license, a sort of a "sticker" that firms can show to prove their customers that they comply with environmental laws.<sup>8</sup> Otherwise, the firm receives a penalty. There are two kinds of firms, compliant and non-compliant, and two kinds of bureaucrats, honest and dishonest. Moreover, we suppose the existence of two crimes: corruption and extortion. Corruption occurs when a dishonest bureaucrat claims a bribe from a compliant firm. Finally, we introduce the existence of an anti-corruption agency that monitors the behaviour of bureaucrats and firms. The agency can be regarded as an exogenous player that affects the choices of the two populations.<sup>9</sup>

When there is no dominant strategy, four dynamic regimes can arise in the economy: two bistable dynamic regimes (i.e. with extreme equilibria and no inner equilibrium) and two with an internal stable equilibrium. In the first two dynamic regimes, similar economies (same rules, same sanctions, etc.) can converge to different stationary states, depending on the initial shares

 $<sup>^{6}</sup>$ See Di Vita (2014) for a discussion of the deterrence effects of sanctions of environmental crimes from the perspective of law and economics.

<sup>&</sup>lt;sup>7</sup>A random-matching approach has been used in several other economic contexts to describe the numerous and heterogeneous situations in which agents interact during pair wise casual meetings. These situations include, among others, the random matching between buyers and sellers in housing transactions, between unemployed workers who look for a job and entrepreneurs who want to hire new workers, or the matching between firms that apply for a loan and banking executives who have to decide whether to grant the loan. See Rogerson and Shimer (2011), Han and Strange (2015) and Dong et al. (2016) for updated surveys and in-depth discussions on the use of random matching models in housing markets, labour markets and credit allocation, respectively.

<sup>&</sup>lt;sup>8</sup>For instance, one can think of ecolabels as an application of the notion of green license described here. See the interesting contribution by Blanco and Lozano (2015) for an evolutionary model examining how the transitional dynamics deriving from certification practices may improve or worsen natural resources.

 $<sup>^{9}</sup>$ See Di Vita (2007) who examines how agents respond to exogenous changes in (dis)incentives to corruption coming from an authority that is not part of the game.

of the strategies in the two populations. In the other two dynamic regimes, instead, the shares of the two populations oscillate around an internal stable equilibrium. This implies that similar economies that differ only for their initial conditions can lie on different trajectories, therefore, they can have different "orbital periods", describing circles of different lengths around the internal equilibrium.

The comparative statics analysis of the dynamic regimes with an internal stable equilibrium shows that policy instruments (sanctions, probability of being discovered by the anti-corruption agency and inspection effort) may reduce both corruption and extortion. The effectiveness of policy instruments depends on their impact on the payoffs of the alternative strategies and on the initial shares of the strategies in the two populations.

Our model differs from other inspection games (both static and evolutionary ones) in several respects. While inspection games, as described above, have a single mixed strategy Nash equilibrium, (therefore, they only allow dynamic regimes with oscillating trajectories), the present game has Nash equilibria both in pure and in mixed strategies, which enriches the dynamics that can arise from the model. Moreover, as it will be shown below, in the present context not only the enforcer's effort but also higher sanctions and detection probability can effectively reduce crimes. In addition, we broaden the possible cases of corruption as compared to other inspection games, in the present model inspectors choose whether to be honest or dishonest rather than their inspection effort. The latter here is captured by the detection probability that encompasses also other factors (e.g. the technologies the inspector has at disposal to perform proper controls). Such probability is initially assumed to be exogenously given in order to focus the analysis on the binary choice (being honest/dishonest) described above, and then properly modified to see the effects of a change in the corresponding parameter on environmental compliance.

The paper is organized as follows. Sections 2 and 3 describe the model, Section 4 shows the basic results, Section 5 deals with the dominance relationship between strategies, Section 6 analyses the dynamic regimes, Section 7 contains the effectiveness of policy instruments, and Section 8 concludes.

## 2 The model

Let us assume that in each instant of time  $t \in [0, +\infty)$ , many pairwise random matchings occur between firms and bureaucrats. Each firm has to choose *ex ante* between two possible strategies: (C) comply with environmental regulations and incur compliance cost  $C_C$ , or (NC) not comply with environmental laws. Each bureaucrat has to choose *ex ante* between two possible strategies: (H) be honest and do her job properly, or (D) be dishonest and accept a bribe from a noncompliant firm or claim a bribe from a compliant firm. Tables 1 and 2 describe the firm's and the bureaucrat's payoff matrix, respectively.

If a compliant firm encounters an honest bureaucrat, it obtains the green license, while if it encounters a dishonest bureaucrat it may be victim of extortion; if that is the case, the compliant firm has to pay an extortion bribe  $(b_e)$ . However, this crime could be discovered by the anticorruption agency with probability  $\theta$ ; in that case, the public administration will compensate the firm for the extortion bribe by offering a refund  $(\eta)$ . If a non-compliant firm encounters an honest bureaucrat, it will be sanctioned  $(s_1)$  for not being compliant with a probability p, that depends on the honest bureaucrat's capacity to discover the firm's violation.<sup>10</sup> Otherwise, if a non-compliant firm encounters a dishonest bureaucrat, it will pay a corruption bribe  $(b_c)$ .

 $<sup>^{10}</sup>$ Such a capacity will obviously depend not only on the bureaucrat's monitoring effort but also on the efficacy of the control instruments that she has at disposal.

Table 1Payoffs of strategies C and NC

	Н	D
C	$\pi_C^H = -C_C$	$\pi^D_C = -C_C - b_e + \theta \eta$
NC	$\pi^H_{NC} = -ps_1$	$\pi^D_{NC} = -b_c - \theta s_2$

Table 2Payoffs of strategies H and D

	C	NC
Н	$\pi_{H}^{C} = w$	$\pi_{H}^{NC} = w$
D	$\pi_D^C = w + b_e - \theta \sigma_1$	$\pi_D^{NC} = w + b_c - \theta \sigma_2$

In this case, it will take the risk of being sanctioned  $(s_2)$  by the anti-corruption agency with probability  $\theta$ . In this case, the firm will be sanctioned both for the corruption crime and for not being compliant with the environmental laws. We suppose that  $C_C > 0$ ,  $b_c > b_e > 0$ ,  $\eta \ge 0$ ,  $s_2 > s_1 > 0, 1 > \theta > 0, 1 > p > 0$ . Let us turn now out attention to the payoff matrix of the bureaucrat. Empirical studies suggest that incentives in wages to bureaucrats may be ineffective in reducing corruption, whereas bribery may be discouraged by the extension of liability rules for the Public Administration and the civil servants (e.g. Di Vita, 2011). For this reason, we will assume that no incentive wage is offered to the honest bureaucrat (i.e. her payoff is independent of the type of firm she encounters) whereas proper sanctions are levied on corrupted bureaucrats. More precisely, if a bureaucrat is honest, she will obtain the wage (w), regardless of the kind of firms she encounters. If a dishonest bureaucrat encounters a compliant firm, she will obtain an extortion bribe in addition to wage, but it will run the risk of being sanctioned ( $\sigma_1$ ) by the anti-corruption agency with probability  $\theta$ .<sup>11</sup> Finally, if a dishonest bureaucrat encounters a noncompliant firm, she will obtain -beyond her wage- the corruption bribe but will take the risk of being sanctioned ( $\sigma_2$ ) by the anti-corruption agency with probability  $\theta$ . We suppose that w > 0,  $b_c > b_e > 0, \, \sigma_1 > 0, \, \sigma_2 > 0, \, 1 > \theta > 0.$ 

## 3 The Dynamics of the game

Let  $c(t) \in [0, 1]$  represent the share of firms adopting strategy C and let  $h(t) \in [0, 1]$  represent the share of bureaucrats adopting strategy H, at time t. Consequently, 1-c(t) and 1-h(t) represent, respectively, the shares of firms playing strategy NC and of bureaucrats playing strategy D.

The firms' expected payoffs from playing strategies C and NC are:

$$\Pi_C(h) = \pi_C^H \cdot h + \pi_C^D \cdot (1-h)$$

<sup>&</sup>lt;sup>11</sup>Notice that a dishonest bureaucrat might hypothetically decide not to ask any extortion bribe ( $b_e = 0$ ) if she meets a compliant firm, in which case she would obviously get no sanction ( $\sigma_1 = 0$ ). In what follows, however, we will focus attention on the more interesting case in which both  $b_e$  and  $\sigma_1$  are strictly positive.

$$\Pi_{NC}(h) = \pi_{NC}^H \cdot h + \pi_{NC}^D \cdot (1-h)$$

where h and 1-h represent the probabilities that a firm is matched with a bureaucrat who plays, respectively, strategy H or D.

The bureaucrats' expected payoffs from playing strategies H and D are:

$$\Pi_H(c) = \pi_H^C \cdot c + \pi_H^{NC} \cdot (1-c)$$
$$\Pi_D(c) = \pi_D^C \cdot c + \pi_D^{NC} \cdot (1-c)$$

where c and 1-c represent the probabilities that a bureaucrat is matched with a firm who plays, respectively, strategy C or NC.

The average payoffs in the population of firms and of bureaucrats are:

$$\overline{\Pi}_F = c \cdot \Pi_C(h) + (1 - c) \cdot \Pi_{NC}(h)$$
$$\overline{\Pi}_B = h \cdot \Pi_H(c) + (1 - h) \cdot \Pi_D(c)$$

We assume that the time evolution of c and h is described by the standard replicator dynamics, a learning-by-imitation model of evolution widely used in economics (see, among others, Hofbauer and Sigmund, 1988; Weibull, 1995). The replicator dynamics postulates that players are bundled rational and update their choices by adopting the relatively more rewarding behaviour that emerges from available observations of others' behaviours. The shares c and h will increase (decrease) the more, the higher (lower) their payoff differential with respect to the population average payoff. Accordingly, in our two-strategy context the dynamic system is:

$$\dot{c} = c[\Pi_C(h) - \overline{\Pi}_F] = c(1-c)[\Pi_C(h) - \Pi_{NC}(h)] \dot{h} = h[\Pi_H(c) - \overline{\Pi}_B] = h(1-h)[\Pi_H(c) - \Pi_D(c)]$$
(1)

where  $\dot{c}$  and  $\dot{h}$  represent the time derivatives dc/dt and dh/dt of the shares c and h, respectively. The factors c(1-c) and h(1-h) are always non-negative, so the signs of  $\dot{c}$  and  $\dot{h}$  will depend respectively on the signs of the payoff differentials.

## 4 Basic results

The system (1) is defined in the unit square S:

$$S = \{ (c,h) \in \mathbb{R}^2 : 0 \le c \le 1, 0 \le h \le 1 \}$$

The graphs of the payoff differentials  $\Pi_C(h) - \Pi_{NC}(h)$  and  $\Pi_H(c) - \Pi_D(c)$  are shown in Figs. 1(a) and 1(b). Strategy C(H) is dominant when the graph of  $\Pi_C(h) - \Pi_{NC}(h)$  ( $\Pi_H(c) - \Pi_D(c)$ ) lies entirely above the *c*-axis (*h*-axis) in the interval [0, 1]. Conversely, strategy NC(D) is dominant when it lies entirely below the *c*-axis (*h*-axis) in the interval [0, 1]. Finally, if it intersects the interior of the interval [0, 1], then no dominant strategy exists. Figs. 1(a) and 1(b) show the possible cases that can be observed.

The payoff differentials can be written as follows:

$$\Pi_C(h) - \Pi_{NC}(h) = b_c - b_e + \theta (\eta + s_2) - C_C - [b_c - b_e + \theta (\eta + s_2) - ps_1] h$$
(2)

$$\Pi_H(c) - \Pi_D(c) = \theta \sigma_2 - b_c + [b_c - b_e + \theta (\sigma_1 - \sigma_2)] c$$
(3)

According to the dynamic system (1),  $\dot{c} = 0$  holds if either c = 0, 1 or if the value of the share h is such that  $\Pi_C(h) - \Pi_{NC}(h) = 0$ , that is:

$$h = \bar{h} := \frac{b_c - b_e + \theta (\eta + s_2) - C_C}{b_c - b_e + \theta (\eta + s_2) - ps_1}$$
(4)

Considering (2), we can distinguish between two cases.

Case (a):

$$b_c - b_e + \theta \ (\eta + s_2) - ps_1 < 0, \ that \ is, \ s_1 > \bar{s}_1$$
 (5)

where  $\bar{s}_1 := \frac{b_c - b_e + \theta (\eta + s_2)}{p}$ . Case (b):

$$b_c - b_e + \theta \ (\eta + s_2) - ps_1 > 0, that \ is, \ s_1 < \bar{s}_1 \tag{6}$$

The graph of the payoff differential  $\Pi_C(h) - \Pi_{NC}(h)$  is a line with positive slope (i.e.,  $\Pi_C(h) - \Pi_{NC}(h)$  is an increasing function of h) in Case (a), while it is a line with negative slope (i.e.,  $\Pi_C(h) - \Pi_{NC}(h)$  is a decreasing function of h) in Case (b). This implies that, in the context of Case (a), the strategy C becomes relatively more remunerative (compared to the strategy NC) when the share of honest bureaucrats h increases; the opposite occurs in Case (b).

Analogously, according to the dynamic system (1), h = 0 holds if either h = 0, 1 or if the value of the share c is such that  $\Pi_H(c) - \Pi_D(c) = 0$ , that is:

$$c = \bar{c} := \frac{b_c - \theta \sigma_2}{b_c - b_e + \theta (\sigma_1 - \sigma_2)} \tag{7}$$

Taking into account (3), we can distinguish between two cases.

Case (c):

$$b_c - b_e + \theta \ (\sigma_1 - \sigma_2) > 0, \ that \ is, \ \sigma_1 > \bar{\sigma}_1 \tag{8}$$

where  $\bar{\sigma}_1 := \sigma_2 - \frac{b_c - b_e}{\theta}$ . Case (d):

$$b_c - b_e + \theta \ (\sigma_1 - \sigma_2) < 0, \ that \ is, \ \sigma_1 < \bar{\sigma}_1 \tag{9}$$

The graph of the payoff differential  $\Pi_H(c) - \Pi_D(c)$  is a line with positive slope (i.e.,  $\Pi_H(c) - \Pi_D(c)$ ) is an increasing function of c) in Case (c), while it is a line with negative slope (i.e.,  $\Pi_H(c) - \Pi_D(c)$ ) is a decreasing function of c) in Case (d). This implies that, in the context of Case (c), the strategy H becomes relatively more remunerative (compared to the strategy D) when the share of compliant firms c increases; the opposite occurs in Case (d).

The four vertices of S, that is (c, h) = (0, 0), (1, 0), (0, 1), (1, 1), are always stationary states of the dynamic system (1). In these stationary states, the populations of firms and bureaucrats play only one strategy. In (1, 1) all firms play C and all bureaucrats play H; in (0, 0) all firms play NC and all bureaucrats play D, and so on.

Another stationary state of the system (1) is the intersection point  $(\bar{c}, \bar{h})$  of the straight lines (4) and (7), when it belongs to the interior of the square S, that is when  $0 < \bar{c} < 1$  and  $0 < \bar{h} < 1$ . At the stationary state  $(\bar{c}, \bar{h})$  all the strategies C, NC, H and D coexist.

Finally, all the points belonging to the side of S with h = 0 (respectively, h = 1) are stationary states in the case in which  $\bar{h} = 0$  (respectively,  $\bar{h} = 1$ ) holds. Analogously, all the points belonging to the side of S with c = 0 (respectively, c = 1) are stationary states if  $\bar{c} = 0$  (respectively,  $\bar{c} = 1$ ) holds.



(a) Increasing payoff differentials.

(b) Decreasing payoff differentials.

#### Fig. 1. Dominance of strategies.

Legend: line (A) dominant strategy C or H, line (B) no dominant strategy, line (C) dominant strategy NC or D; line (D) dominant strategy C or H, line (E) no dominant strategy, line (F) dominant strategy NC or D.

## 5 Dominance relationship

In this section, we define the conditions under which a given strategy does not dominate the alternative one, in each population of players. Proposition 1 refers to Case (a) and Case (b), while Proposition 2 refers to Case (c) and Case (d).

**Proposition 1** In Case (a) (see (5)),  $\Pi_C(h) - \Pi_{NC}(h)$  is strictly increasing in h (see Fig. 1(a)), and there is no dominance of strategies if:

$$s_2 < \bar{s}_2 := \frac{C_C + b_e - b_c}{\theta} - \eta \quad and \quad s_1 > \frac{C_C}{p} \tag{10}$$

In Case (b) (see (6)),  $\Pi_C(h) - \Pi_{NC}(h)$  is strictly decreasing in h (see Fig. 1(b)), and there is no dominance of strategies if:

$$s_2 > \bar{s}_2 \quad and \quad s_1 < \frac{C_C}{p}$$
 (11)

#### **Proof.** See Appendix.

In Case (a), if condition (10) holds, then no strategy dominates the other one, and the graph of the payoff differential  $\Pi_C(h) - \Pi_{NC}(h)$  intersects the *h*-axis at  $h = \bar{h} \in (0, 1)$ . In this case, for  $h > \bar{h}$  (respectively,  $h < \bar{h}$ ), it holds  $\Pi_C(h) - \Pi_{NC}(h) > 0$  (respectively,  $\Pi_C(h) - \Pi_{NC}(h) < 0$ ) (see (4)).

On the contrary, in Case (b), if condition (11) holds, then no strategy dominates the other one, and the graph of the payoff differential  $\Pi_C(h) - \Pi_{NC}(h)$  intersects the *h*-axis at  $h = \bar{h} \in$ (0,1). If so, for  $h > \bar{h}$  (respectively,  $h < \bar{h}$ ), it holds  $\Pi_C(h) - \Pi_{NC}(h) < 0$  (respectively,  $\Pi_C(h) - \Pi_{NC}(h) > 0$ ) (see (4)). **Proposition 2** In Case (c) (see (8)),  $\Pi_H(c) - \Pi_D(c)$  is strictly increasing (see Fig. 1(a)), and there is no dominance of strategies if:

$$\sigma_1 > \frac{b_e}{\theta} \quad and \quad \sigma_2 < \frac{b_c}{\theta}$$
 (12)

In Case (d) (see (9)),  $\Pi_H(c) - \Pi_D(c)$  is strictly decreasing (see Fig. 1(b)), and there is no dominance of strategies if:

$$\sigma_1 < \frac{b_e}{\theta} \quad and \quad \sigma_2 > \frac{b_c}{\theta}$$
 (13)

**Proof.** See Appendix.

In Case (c) if condition (12) holds, then no strategy dominates the other one, and the graph of the payoff differential  $\Pi_H(c) - \Pi_D(c)$  intersects the c-axis at  $c = \bar{c} \in (0, 1)$ . If this is the case for  $c > \bar{c}$  (respectively,  $c < \bar{c}$ ), it holds  $\Pi_H(c) - \Pi_D(c) > 0$  (respectively,  $\Pi_H(c) - \Pi_D(c) < 0$ ) (see (7)).

The same result (no dominant strategy) occurs in Case (d) if condition (13) holds, but in this case the graph of the payoff differential  $\Pi_H(c) - \Pi_D(c)$  intersects the *c*-axis at  $c = \bar{c} \in (0, 1)$  from above: for  $c > \bar{c}$  (respectively,  $c < \bar{c}$ ), it holds  $\Pi_H(c) - \Pi_D(c) < 0$  (respectively,  $\Pi_H(c) - \Pi_D(c) > 0$ ) (see (7)).

## 6 Dynamic regimes

When a dominated strategy exists, the share of agents adopting it decreases monotonically over time and approaches (asymptotically) the value 0; therefore, in such a context, the dynamics is very simple. The most interesting dynamic regimes arise when no strategy dominates the other one, in each population of agents. In such cases, the internal stationary state  $(\bar{c}, \bar{h})$  exists. The following subsections illustrate these dynamic regimes. The classification of regimes that will be given below provides an exhaustive description all the possible cases that can occur:<sup>12</sup>

- 1) the case in which the parameter values satisfy the conditions (5) and (8) characterizing, respectively, Case (a) (relatively to the population of firms) and Case (c) (relatively to the population of bureaucrats);
- 2) the case in which the parameter values satisfy the conditions (6) and (9) characterizing, respectively, Case (b) and Case (d);
- 3) the case in which the parameter values satisfy the conditions (6) and (8) characterizing, respectively, Case (b) and Case (c);
- 4) the case in which the parameter values satisfy the conditions (5) and (9) characterizing, respectively, Case (a) and Case (d).

As it will be shown below, the first two cases are characterized by bi-stable dynamics, while the latter two by oscillatory dynamics.

The proofs of the following propositions are straightforward, since the dynamic regimes that may be observed under replicator equations, in a context with two populations and two strategies, have been completely classified (see Hofbauer and Sigmund, 1988).

 $<sup>^{12}</sup>$  For simplicity, we do not consider the non-robust cases with  $s_1=\bar{s}_1$  and/or  $\sigma_1=\bar{\sigma}_1$ 

#### 6.1 Bi-stable dynamics

#### 6.1.1 Dynamic regime in the context of Cases (a) and (c)

This case is characterized by the conditions (see (5) and (8)):

$$s_1 > \frac{b_c - b_e + \theta \left(\eta + s_2\right)}{p} \tag{14}$$

$$\sigma_1 > \sigma_2 - \frac{b_c - b_e}{\theta} \tag{15}$$

which, as shown in Section 5, imply that the payoff differentials  $\Pi_C(h) - \Pi_{NC}(h)$  and  $\Pi_H(c) - \Pi_D(c)$  are strictly increasing in h and c, respectively. Furthermore, in such a context, no dominance relationship (between the two available strategies) exists, in each population, if the following conditions are satisfied (see (10) and (12)):

$$s_2 < \frac{C_C + b_e - b_c}{\theta} - \eta \quad and \quad s_1 > \frac{C_C}{p} \tag{16}$$

$$\sigma_1 > \frac{b_e}{\theta} \quad and \quad \sigma_2 < \frac{b_c}{\theta}$$
(17)

Notice that if condition (17) holds, then also condition (15) holds. If conditions (14), (16) and (17) are satisfied, then a "bi-stable" dynamic regime is observed, described by the following proposition:

**Proposition 3** If conditions (14), (16) and (17) are satisfied, then the stationary states (c, h) = (0,0) and (c,h) = (1,1) are sinks (i.e., locally attractive), the stationary states (c,h) = (1,0) and (c,h) = (0,1) are sources (i.e., repulsive) and the stationary state  $(c,h) = (\bar{c},\bar{h})$ , in the interior of the square S, is a saddle point. The basins of attraction of (0,0) and (1,1) are separated by the stable branch of  $(\bar{c},\bar{h})$  (see Fig. 2(a)).

In such a context, strategy C is the best reply when the share of honest bureaucrats is high, while strategy NC is the best reply when the share of honest bureaucrats is low.<sup>13</sup> With regard to bureaucrats' behaviour, instead, strategy H is the best reply when the share of compliant firms is high, while strategy D is the best reply when the share of the compliant firms is low. This occurs because:

- If the share of honest bureaucrats is high, for firms is more rewarding to adopt strategy C, since  $s_1$  and p are relatively high, while  $C_C$  is relatively low  $(s_1 > C_C/p)$ .
- If the share of honest bureaucrats is low, for firms is more rewarding to adopt strategy NC, since  $s_2$ ,  $b_c$ ,  $\theta$  and  $\eta$  are relatively low, while  $C_C$  and  $b_e$  are relatively high ( $s_2 < (C_C + b_e b_c)/\theta \eta$ ).
- If the share of compliant firms is high, for bureaucrats is more rewarding to adopt strategy H, since  $\sigma_1$  and  $\theta$  are relatively high, while  $b_e$  is relatively low  $(\sigma_1 > b_e/\theta)$ .
- If the share of compliant firms is low, for bureaucrats is more rewarding to adopt strategy D, since  $\sigma_2$  and  $\theta$  are relatively low, while  $b_c$  is relatively high ( $\sigma_2 < b_c/\theta$ ).

<sup>&</sup>lt;sup>13</sup>This can be easily shown by observing line (B) in Fig. 1(a) that describes Case (a): indeed, when h is sufficiently high,  $\Pi_C(h) - \Pi_{NC}(h)$  is positive, therefore the strategy C is the best response of firms to the bureaucrats' behaviour as it is more remunerative than strategy NC. Mutatis mutandis, the same reasoning underlies the identification of the best replies in all the other cases that will be described below.

An economy can converge to the "vicious" stationary state (0,0) - in which all firms are non-compliant and all bureaucrats are dishonest - if both initial shares of compliant firms and honest bureaucrats are relatively low. In point (0,0) corruption is the only existing crime. On the contrary, an economy can converge to the "virtuous" stationary state (1,1) - in which all firms are compliant and all bureaucrats are honest - if both initial shares of compliant firms and honest bureaucrats are relatively high. In point (1,1) there are no crimes.

#### 6.1.2 Dynamic regime in the context of Cases (b) and (d)

This case is characterized by the conditions (see (6) and (9)):

$$s_1 < \frac{b_c - b_e + \theta \left(\eta + s_2\right)}{p} \tag{18}$$

$$\sigma_1 < \sigma_2 - \frac{b_c - b_e}{\theta} \tag{19}$$

which imply that the payoff differentials  $\Pi_C(h) - \Pi_{NC}(h)$  and  $\Pi_H(c) - \Pi_D(c)$  are strictly decreasing in h and c, respectively. Furthermore, in such a context, no dominance relationship exists, in each population, if the following conditions are satisfied (see (11)) and (13)):

$$s_2 > \frac{C_C + b_e - b_c}{\theta} - \eta \quad and \quad s_1 < \frac{C_C}{p} \tag{20}$$

$$\sigma_1 < \frac{b_e}{\theta} \quad and \quad \sigma_2 > \frac{b_c}{\theta}$$
(21)

Notice that if condition (21) holds, then also condition (19) holds. If conditions (18), (20) and (21) are satisfied, the bi-stable regime described by the following proposition occurs:

**Proposition 4** If the condition (18), (20) and (21) are satisfied, then the stationary states (c,h) = (0,1) and (c,h) = (1,0) are sinks, the stationary states (c,h) = (0,0) and (c,h) = (1,1) are sources and the stationary state  $(c,h) = (\bar{c},\bar{h})$ , in the interior of the square S, is a saddle point. The basins of attraction of (0,1) and (1,0) are separated by the stable branch of  $(\bar{c},\bar{h})$  (see Fig. 2(b)).

In such a context, strategy C is the best reply when the share of honest bureaucrats is low, while strategy NC is the best reply when the share of honest bureaucrats is high. With regard to bureaucrats' behaviour, instead, strategy H is the best reply when the share of compliant firms is low, while strategy D is the best reply when the share of the compliant firms is high. This occurs because:

- If the share of honest bureaucrats is high, for firms is more rewarding to adopt strategy NC, since  $s_1$  and p are relatively low, while  $C_C$  is relatively high  $(s_1 < C_C/p)$ .
- If the share of honest bureaucrats is low, for firms is more rewarding to adopt strategy C, since  $s_2$ ,  $b_c$ ,  $\theta$  and  $\eta$  are relatively high, while  $C_C$  and  $b_e$  are relatively low ( $s_2 > (C_C + b_e b_c)/\theta \eta$ ).
- If the share of compliant firms is high, for bureaucrats is more rewarding to adopt strategy D, since  $\sigma_1$  and  $\theta$  are relatively low, while  $b_e$  is relatively high  $(\sigma_1 < b_e/\theta)$ .
- If the share of compliant firms is low, for bureaucrats is more rewarding adopt strategy H, since  $\sigma_2$  and  $\theta$  are relatively high, while  $b_c$  is relatively low ( $\sigma_2 > b_c/\theta$ ).





(a) Cases (a) and (c). Parameter values:  $C_C = 129.09, b_c = 75, b_e = 50, p = 0.67, \theta = 0.29, C_C = 156.808, b_c = 75, b_e = 50, p = 0.63, b_c = 129.09, b_c = 129.09,$  $\eta = 95.76, s_1 = 217.110, s_2 = 202.583,$  $\sigma_1 = 231.191, \ \sigma_2 = 193.019.$ 

(b) Cases (b) and (d). Parameter values:  $\theta = 0.22, \ \eta = 275.35, \ s_1 = 144.174, \ s_2 = 610.660,$  $\sigma_1 = 13.209, \ \sigma_2 = 546.598.$ 

Fig. 2. Path-dependent dynamics. Legend: • attractors, • repellers.

An economy can converge to the stationary state (0,1) - in which no firm is compliant but all bureaucrats are honest - if the initial share of compliant firms is relatively low and the initial share of honest bureaucrats is relatively high. In point (0,1) there are no crimes.<sup>14</sup> On the contrary, an economy can converge to the stationary state (1,0) - in which all firms are compliant and all bureaucrats are dishonest - if the initial share of compliant firms is relatively high and the initial share of honest bureaucrats is relatively low. In point (1,0) extortion is the only existing crime.

The states (c,h) = (0,0), (0,1), (1,0), (1,1), when locally attractive, as in Propositions 3 and 4, are Nash equilibria. This finding follows from standard results in evolutionary game theory (see, e.g., Weibull, 1995). We can interpret Nash equilibria as social conventions, that is, as customary and expected states of things in which no single individual has an incentive to modify her choices if the others do not modify theirs.

Performing numerical simulations in the context of the bi-stable dynamics examined above, it is possible to show the dynamics that may emerge in the model from an increase in the policy parameters, namely, the parameters  $(\theta, s_1, s_2, \sigma_1, \sigma_2)$  of the model that the policy-maker can directly modify. Let us first consider the case in which the "virtuous" equilibrium (1,1) and the "vicious" equilibrium (0,0) are the two attractors of the system (Cases (a) and (c) above). As Figs. 3(a) and 3(b) show, an increase in the effort of the anti-corruption agency that increases

 $<sup>^{14}</sup>$ By this we mean that at this stationary state there is neither corruption nor extortion. There is, however, a widespread violation of the environmental laws since all firms are non-compliant, but this will be properly sanctioned by bureaucrats (that are all honest), provided they manage to discover the violation.



Fig. 3. Basins of attraction of Cases (a) and (c): low and high value of  $\theta$ . Parameter values:  $C_C = 100$ ,  $b_c = 75$ ,  $b_e = 50$ ,  $\eta = 50$ , p = 0.5,  $s_1 = 225$ ,  $s_2 = 25$ ,  $\sigma_1 = 100$ ,  $\sigma_2 = 50$ . Legend: • attractors, • repellers.

the probability  $\theta$  of discovering crimes will tend to enhance the attraction basin of the virtuous equilibrium (1, 1) and to reduce that of the vicious equilibrium (0, 0). Stated differently, in this case an increase in  $\theta$  will increase the range of initial values (c, h) that make the system eventually converge towards the "ideal" equilibrium in which all firms are compliant and all bureaucrats are honest. As a consequence, this will automatically also decrease the range of initial values (c, h)that lead to the opposite undesirable outcome in which all firms are non-compliant and all bureaucrats dishonest. Figs. 4(a) and 4(b) show the effects of an increase in the sanctions  $s_1$  and  $s_2$  levied by the Public Administration on non-compliant firms. Even in this case, an increase in  $s_1$  and  $s_2$  tends to enlarge the attraction basin of (1, 1) and conversely to reduce that of (0, 0), increasing the range of values (c, h) that lead at the end of the day to the desirable situation in which none violates the laws. Notice that, as Figs. 4(a) and 4(b) shows, an increase in the sanctions level does not modify the share of compliant firms  $\bar{c}$  at the (unstable) inner equilibrium  $(\bar{c}, \bar{h})$ .

Let us now focus attention on the basins of attraction of Cases (b) and (d) above (cf. Figs. 5(a) and 5(b) and Figs. 6(a) and 6(b)). In this case, the extreme equilibria (1,1) and (0,0) are repellors and cannot be achieved, while the two attractors are now (1,0) and (0,1) in which one entire population respects the law whereas the others violate it. In this case, an increase in the probability  $\theta$  that the anti-corruption agency may discover the crime increases the share of compliant firms  $\bar{c}$  and of honest bureaucrats  $\bar{h}$  at the inner equilibrium (Figs. 5(a) and 5(b)). Instead, an increase in the sanctions  $\sigma_1$  and  $\sigma_2$  (imposed on the bureaucrats for the extortion and corruption crimes, respectively) increases the share of compliant firms  $\bar{c}$  but it does not affect that of honest bureaucrats  $\bar{h}$  at the inner equilibrium (Figs. 6(a) and 6(b)). Moreover, an increase in  $\sigma_1$  and  $\sigma_2$  augments the attraction basin of (0, 1) in which all firms are non-compliant but all bureaucrats are honest; vice versa it shrinks the attraction basin of (1,0) in which all firms are compliant while all bureaucrats are dishonest. This implies that higher sanctions on the bureaucrats tend to increase the probability that the system may eventually end up in the equilibrium (0,1) in which no extortion and/or corruption takes place (since all bureaucrats



Fig. 4. Basins of attraction of Cases (a) and (c): low and high values of sanctions  $s_1$  and  $s_2$ . Parameter values:  $C_C = 100$ ,  $b_c = 75$ ,  $b_e = 50$ ,  $\eta = 50$ , p = 0.5,  $\theta = 0.66$ ,  $\sigma_1 = 100$ ,  $\sigma_2 = 50$ . Legend: • attractors,  $\circ$  repellers.

are honest), while they tend to decrease the likelihood of ending up in the equilibrium (1,0) characterized by the maximum number of extortion crimes (since all bureaucrats are dishonest while no firm tries to corrupt them).

## 6.2 Oscillatory dynamics

#### 6.2.1 Dynamic regime in the context of Cases (b) and (c)

This context is characterized by the following conditions (see (6) and (8)):

$$s_1 < \frac{b_c - b_e + \theta \left(\eta + s_2\right)}{n} \tag{22}$$

$$\sigma_1 > \sigma_2 - \frac{b_c - b_e}{\theta} \tag{23}$$

which imply that the payoff differentials  $\Pi_C(h) - \Pi_{NC}(h)$  and  $\Pi_H(c) - \Pi_D(c)$  are, respectively, strictly decreasing in h and strictly increasing in c. In such a context, no dominance relationship exists, in each population, if the following conditions are satisfied (see (11) and (12)):

$$s_2 > \frac{C_C + b_e - b_c}{\theta} - \eta \quad and \quad s_1 < \frac{C_C}{p}$$

$$\tag{24}$$

$$\sigma_1 > \frac{b_e}{\theta} \quad and \quad \sigma_2 < \frac{b_c}{\theta}$$

$$\tag{25}$$

Notice that if condition (25) holds, then also condition (23) holds. The following proposition illustrates the basic properties of the dynamic regime observed if conditions (22), (24) and (25) are satisfied:



Fig. 5. Basins of attraction of Cases (b) and (d). Parameter values:  $C_C = 100, b_c = 75, b_e = 50, \eta = 50, p = 0.5, s_1 = 20, s_2 = 2000, \sigma_1 = 20, \sigma_2 = 2000.$ Legend: • attractors, • repellers.

**Proposition 5** If conditions (22), (24) and (25) are satisfied, then the stationary states (c, h) = (0, 1), (1, 0), (0, 0) and (1, 1) are saddle points, while the internal stationary state  $(c, h) = (\bar{c}, \bar{h})$  is a (Lyapunov) stable stationary state surrounded by closed trajectories turning counter-clockwise (see Fig. 7(a)).

In such a context, strategy C is the best reply when the share of honest bureaucrats is low, while strategy NC is the best reply when the share of honest bureaucrats is high. With regard to bureaucrats' behaviour, instead, strategy H is the best reply when the share of compliant firms is high, while strategy D is the best reply when the share of the compliant firms is low. This occurs because:

- If the share of honest bureaucrats is high, for firms is more rewarding to adopt strategy NC, since  $s_1$  and p are relatively low, while  $C_C$  is relatively high  $(s_1 < C_C/p)$ .
- If the share of honest bureaucrats is low, for firms is more rewarding to adopt strategy C, since  $s_2$ ,  $b_c$ ,  $\theta$  and  $\eta$  are relatively high, while  $C_C$  and  $b_e$  are relatively low ( $s_2 > (C_C + b_e b_c)/\theta \eta$ ).
- If the share of compliant firms is high, for bureaucrats is more rewarding to adopt strategy H, since  $\sigma_1$  and  $\theta$  are relatively high, while  $b_e$  is relatively low  $(\sigma_1 > b_e/\theta)$ .
- If the share of compliant firms is low, for bureaucrats is more rewarding to adopt strategy D, since σ<sub>2</sub> and θ are relatively low, while b<sub>c</sub> is relatively high (σ<sub>2</sub> < b<sub>c</sub>/θ).

This oscillatory dynamics could be explained using the *prey-predator* conceptual framework à la Lotka-Volterra that is commonly used in evolutionary biology and is largely adopted also in the environmental economics literature (e.g. Brander and Taylor, 1998; Rosser, 2011; Antoci et al., 2005, 2016). In the present context, dishonest bureaucrats play the role of the predators and non-compliant firms are the preys. In fact, a high share of non-compliant firm "attracts" dishonest bureaucrats (just like preys attract predators) for the possibility that the latter may



Fig. 6. Basins of attraction of Cases (b) and (d). Parameter values:  $C_C = 100, b_c = 75, b_e = 50, \eta = 50, p = 0.5, \theta = 0.33, s_1 = 20, s_2 = 2000.$ Legend: • attractors, • repellers.

obtain a corruption bribe from the former. Starting from an initial condition in which the share of Not Complaint firms is high (many preys), for bureaucrats is more rewarding to adopt strategy D, therefore the share of predators increases. However, the increase of dishonest bureaucrats decreases the share of preys since the firms' best reply is to adopt strategy C. A reduction in the number of preys decreases the share of predators, since the bureaucrats' best reply is to adopt strategy H when the share of compliant firms is high. A lower share of predators allows the proliferation of the preys: for firms it becomes more rewarding to adopt strategy NC if the share of honest bureaucrats is high. And so on, leading to the oscillatory dynamics described above.

#### 6.2.2 Dynamic regime in the context of Cases (a) and (d)

This context is characterized by the following conditions (see (5) and (9)):

$$s_1 > \frac{b_c - b_e + \theta \ (\eta + s_2)}{p}$$
(26)

$$\sigma_1 < \sigma_2 - \frac{b_c - b_e}{\theta} \tag{27}$$

which imply that the payoff differentials  $\Pi_C(h) - \Pi_{NC}(h)$  and  $\Pi_H(c) - \Pi_D(c)$  are, respectively, strictly increasing in h and strictly decreasing in c. Furthermore, in such a context, no dominance relationship exists, in each population, if the following conditions are satisfied (see (10) and (13)):

$$s_2 < \frac{C_C + b_e - b_c}{\theta} - \eta \quad and \quad s_1 > \frac{C_C}{p} \tag{28}$$

$$\sigma_1 < \frac{b_e}{\theta} \quad and \quad \sigma_2 > \frac{b_c}{\theta}$$
(29)

Notice that if condition (29) holds, then also condition (27) holds. The following proposition illustrates the basic properties of the dynamic regime observed if conditions (26), (28) and (29) are satisfied:



(a) Cases (b) and (c). Parameter values:  $C_C = 129.069, b_c = 75, b_e = 50, p = 0.67,$   $\theta = 0.29, \eta = 95.76, s_1 = 72.248, s_2 = 531.428,$  $\sigma_1 = 72.248, \sigma_2 = 182.513.$ 

(b) Cases (a) and (d). Parameter values:  $C_C = 111.367, b_c = 75, b_e = 50, p = 0.78,$   $\theta = 0.09, \eta = 123.47, s_1 = 205.290, s_2 = 252.576,$  $\sigma_1 = 8.847, \sigma_2 = 1272.762.$ 

Fig. 7. Dynamics with a stable internal equilibrium. Legend:  $\Box$  saddle points.

**Proposition 6** If conditions (26), (28) and (29) are satisfied, then the stationary states (c, h) = (0, 1), (1, 0), (0, 0) and (1, 1) are saddle points, while the internal stationary state  $(c, h) = (\bar{c}, \bar{h})$  is a (Lyapunov) stable stationary state surrounded by closed trajectories turning clockwise (see Fig. 7(b)).

In such a context, strategy C is the best reply when the share of honest bureaucrats is high, while strategy NC is the best reply when the share of honest bureaucrats is low. With regard to bureaucrats' behaviour, instead, strategy H is the best reply when the share of compliant firms is low, while strategy D is the best reply when the share of the compliant firms is high. This occurs because:

- If the share of honest bureaucrats is high, for firms is more rewarding to adopt strategy C, since  $s_1$  and p are relatively high, while  $C_C$  is relatively low  $(s_1 > C_C/p)$ .
- If the share of honest bureaucrats is low, for firms is more rewarding to adopt strategy NC, since  $s_2$ ,  $b_c$ ,  $\theta$  and  $\eta$  are relatively low, while  $C_C$  and  $b_e$  are relatively high ( $s_2 < (C_C + b_e b_c)/\theta \eta$ ).
- If the share of compliant firms is high, for bureaucrats is more rewarding to adopt strategy D, since  $\sigma_1$  and  $\theta$  are relatively low, while  $b_e$  is relatively high  $(\sigma_1 < b_e/\theta)$ .
- If the share of compliant firms is low, for bureaucrats is more rewarding to adopt strategy H, since  $\sigma_2$  and  $\theta$  are relatively high, while  $b_c$  is relatively low ( $\sigma_2 > b_c/\theta$ ).

Analogously to what has been done before, the *prey-predator* conceptual framework can be used also in the present context to explain the oscillatory dynamics observed here. For this purpose, the dishonest bureaucrats can still be interpreted as the predators, whereas the role of the preys is now played by the compliant firms. In fact, if bureaucrats expect that the probability of being discovered (and sanctioned) by the anti-corruption agency is relatively low, a high share of compliant firms will attract a high share of dishonest bureaucrats. Starting from an initial condition in which the share of compliant firms is high (many preys), therefore, the share of predators increases since bureaucrats find more rewarding to adopt strategy D. However, the increase of dishonest bureaucrats reduces the share of preys: the best reply for the firms is to adopt strategy NC. A lower number of preys decreases the share of predators, since the best reply for bureaucrats is to adopt strategy H if the share of non-compliant firms is high. A lower share of predators allows the proliferation of the preys: for firms is more rewarding to adopt strategy C if the share of honest bureaucrats is high. And so on.

The state  $(c, h) = (\bar{c}, \bar{h})$ , in Propositions 5 and 6, corresponds to the mixed-strategy Nash equilibrium of the one shot (static) game defined by the payoff matrices shown in Tables 1 and 2. Accordingly, the firm chooses the strategy C with probability  $\bar{c}$  and the bureaucrat chooses the strategy H with probability  $\bar{h}$ ; therefore  $(\bar{c}, \bar{h})$  can be interpreted as the equilibrium that would be achieved if all agents were perfectly rational. Given that individuals are assumed here to be boundedly rational, in the present context the system oscillates around the equilibrium  $(\bar{c}, \bar{h})$ . The latter, therefore, can be interpreted as the time-average values of the shares of compliant firms and honest bureaucrats, evaluated over the closed trajectories in Figs. 7(a) and 7(b). In this sense  $(\bar{c}, \bar{h})$  can estimate the behaviour of economic agents in random observations over long time periods (see Weibull, 1995).

## 7 Comparative statics

This section studies the effects that variations in the parameter values can have on the coordinates of the internal equilibrium  $(\bar{c}, \bar{h})$ . We focus our analysis on the two dynamic regimes in which the internal equilibrium  $(\bar{c}, \bar{h})$  is stable, therefore, the values of c and h represent also the average values along the closed trajectories (i.e., the context of Cases (b) and (c), and that of Cases (a) and (d)). In the other cases, the internal equilibrium is a saddle point, therefore, it is not stable. The following Proposition 7 provides the general comparative statics results concerning variations in the parameters of the model whose values can be influenced by the Public Administration's choices:

- p: probability for a non-compliant firm to be discovered by a honest bureaucrat;
- $\theta$ : probability for a non-compliant firm (that has encountered a dishonest bureaucrat who did not sanction it) to be discovered by the anti-corruption agency;
- $s_1$ : sanction incurred by the firm for not being compliant;
- $s_2$ : sanction incurred by the firm for its corruption crime and for not being compliant;
- $\sigma_1$ : sanction incurred by the bureaucrat for the extortion crime;
- $\sigma_2$ : sanction incurred by the bureaucrat for the corruption crime.

The symbols  $x \uparrow$  and  $x \downarrow$  indicate, respectively, that the value of x increases or decreases, where x may represent  $\bar{c}$ ,  $\bar{h}$ , or a parameter of the model.

#### **Proposition 7**

- 1) If  $p \uparrow$ , then  $\bar{c}$  remains always constant, while  $\bar{h} \uparrow if$  and only if (iff)  $s_2 > (C_C + b_e b_c)/\theta \eta$ .
- 2) If  $\theta \uparrow$ , then  $\bar{c} \uparrow iff \sigma_2 b_e > \sigma_1 b_c$ , while  $\bar{h} \uparrow iff s_1 < C_C/p$ .
- 3) If  $s_1 \uparrow$ , then  $\bar{c}$  remains always constant, while  $\bar{h} \uparrow iff s_2 > (C_C + b_e b_c)/\theta \eta$ .
- 4) If  $s_2 \uparrow$ , then  $\bar{c}$  remains always constant, while  $\bar{h} \uparrow iff s_1 < C_C/p$ .
- 5) If  $\sigma_1 \uparrow$ , then  $\bar{c} \uparrow$  iff  $\sigma_2 > b_c/\theta$ , while  $\bar{h}$  remains always constant.
- 6) If  $\sigma_2 \uparrow$ , then  $\bar{c} \uparrow$  iff  $\sigma_1 < b_e/\theta$ , while  $\bar{h}$  remains always constant.

**Proof.** This can be easily proved by looking at the signs of the partial derivatives of functions (4) and (7).  $\Box$ 

### 7.1 Comparative statics in the context of Cases (b) and (c)

When Cases (b) and (c) occur, it is important to recall that the following properties hold:

- The payoff differential  $\Pi_C(h) \Pi_{NC}(h)$  is a decreasing function of h, therefore, the higher is the share of honest bureaucrats h, the lower is the payoff of strategy C relative to strategy NC. This implies that, for sufficiently low values of h, it holds  $\Pi_C(h) - \Pi_{NC}(h) > 0$ (therefore,  $\dot{c} > 0$ ), while the opposite occurs for sufficiently high values of h.
- The payoff differential  $\Pi_H(c) \Pi_D(c)$  is an increasing function of c, therefore, the higher is the share of compliant firms c, the higher is the payoff of strategy H relative to strategy D. It follows that, when c is sufficiently low, it holds  $\Pi_H(c) - \Pi_D(c) < 0$  (therefore,  $\dot{h} < 0$ ), while the opposite occurs when c is sufficiently high.

In such a context, as described in Section 6, in an *environment* characterized by a low share of honest bureaucrats and of compliant firms, C is the best reply of the firms, while D is the best reply of the bureaucrats. In other words, from conditions (24) and (25) it follows that firms adopting strategy C show an *evolutionary advantage* over firms adopting the alternative strategy NC. The same applies to bureaucrats that adopt strategy D over those adopting strategy H. If the Public Administration increases the sanctions  $(s_1, s_2, \sigma_1, \text{ and } \sigma_2)$  and its monitoring effort so as to increase the probabilities of discovering non-compliant firms  $(p \text{ and } \theta)$ , the effects will be mixed (see Table 3). In fact, an increase in the policy parameters  $(p, \theta, s_1, s_2, \sigma_1, \sigma_2)$  has the effect of increasing (or at least not decreasing) the share of honest bureaucrats (i.e.,  $\bar{h}$ ), but decreasing (or at least not increasing) the share of compliant firms (i.e.,  $\bar{c}$ ).

#### 7.2 Comparative statics in the context of Cases (a) and (d)

When Cases (a) and (d) occur, the following properties hold:

- The payoff differential  $\Pi_C(h) \Pi_{NC}(h)$  is an increasing function of h, so that strategy C becomes more and more rewarding with respect to strategy NC as the share of honest bureaucrats h increases. It follows that  $\Pi_C(h) \Pi_{NC}(h) < 0$  (therefore,  $\dot{c} < 0$ ) when h is low enough, while the opposite occurs when h is high enough.
- The payoff differential  $\Pi_H(c) \Pi_D(c)$  is a decreasing function of c, therefore, the higher is the share of compliant firms c, the lower is the payoff of strategy H with respect to strategy D. As a consequence, it holds  $\Pi_H(c) - \Pi_D(c) > 0$  (therefore,  $\dot{h} > 0$ ) for relatively low values of c (when  $c < \bar{c}$ ), while the opposite occurs for relatively high values of c (when  $c > \bar{c}$ ).

	$\bar{c}$	$ar{h}$
p	_	$\uparrow$
heta	$\downarrow$	$\uparrow$
$s_1$	_	$\uparrow$
82	_	$\uparrow$
$\sigma_1$	$\downarrow$	_
$\sigma_2$	$\downarrow$	_

Table 3 Cases (b) and (c): monotonic relations between equilibrium shares ( $\bar{c}, \bar{h}$ ) and their parameters

Legend:  $\uparrow$  Increasing,  $\downarrow$  Decreasing, - Independent.

#### Table 4

Cases (a) and (d): monotonic relations between equilibrium shares  $(\bar{c}, \bar{h})$  and their parameters

	$\bar{c}$	$ar{h}$
p	-	$\downarrow$
heta	$\uparrow$	$\downarrow$
$s_1$	-	$\downarrow$
$s_2$	_	$\downarrow$
$\sigma_1$	$\uparrow$	_
$\sigma_2$	$\uparrow$	_

Legend:  $\uparrow$  Increasing,  $\downarrow$  Decreasing, - Independent.

In such a context, as described in Section 6, in an *environment* characterized by a low share of honest bureaucrats and a low share of compliant firms, then NC turns out to be the best reply of the firms, while H is the best reply of the bureaucrats. From conditions (28) and (29), we have that firms adopting strategy NC and bureaucrats adopting strategy H show an *evolutionary advantage* over the member of the two populations who adopted the alternative strategies (C and D, respectively). If there exist high shares of non-compliant firms and dishonest bureaucrats, and the Public Administration tries to address the problem by increasing sanctions ( $s_1$ ,  $s_2$ ,  $\sigma_1$ , and  $\sigma_2$ ) and the monitoring effort so as to enhance the probabilities p and  $\theta$ , comparative static analysis shows that the effects will be mixed (see Table 4). In fact, an increase in the policy parameters (p,  $\theta$ ,  $s_1$ ,  $s_2$ ,  $\sigma_1$ ,  $\sigma_2$ ) increases (or at least it does not decrease) the share of compliant firms (i.e.,  $\bar{c}$ ), but, at same time, it tends to reduce (or at least it does not increase) the share of honest bureaucrats (i.e.,  $\bar{h}$ ).

## 8 Conclusions

Environmental corruption has recently attracted particular attention among the media and policy-makers due to the discovery of a few important frauds and scandals that may have large

implications on the credibility of environmental data and policies. To contribute to the current debate on environmental corruption, this paper investigates it from a novel perspective, adopting a theoretical framework (i.e. the random matching models) that has been used in other contexts of economic theory (e.g. labour economics) but has found little or no application among the studies on environmental corruption so far. For this purpose, this paper proposes a random matching evolutionary game between a population of firms and a population of bureaucrats. In each encounter a bureaucrat checks the firm's compliance with environmental regulations. When the firm respect the environmental laws, it obtains a "green" license; otherwise, it receives a penalty. We assume the existence of two types of firms, compliant and non-compliant, two types of bureaucrats, honest and dishonest, and also two types of crimes, corruption (when a dishonest bureaucrat accepts a bribe from a non-compliant firm), and extortion (when a dishonest bureaucrat extorts a bribe from a compliant firm).

Four possible dynamic regimes emerge from the analysis of the model: two of them are bistable, while the other two have an internal stable equilibrium, which corresponds to the mixed-strategy Nash equilibrium of the one-shot static game, surrounded by closed trajectories. In the two bistable regimes, the dynamics is path-dependent and the outcome depends on the initial conditions, i.e., the share of compliant firms and of honest bureaucrats that characterize the economy at the beginning. In the two bistable regimes, moreover, all agents within each population end up adopting the same strategy at the equilibrium. In the first regime one equilibrium is "virtuous", all firms are compliant and all bureaucrats are honest, while the other is "vicious", all firms are non-compliant and all bureaucrats are dishonest. In the second regime, instead, one of the two population behaves in a virtuous way while the other does not: in one equilibrium all firms are compliant but all bureaucrats are dishonest, while in the other all bureaucrats are honest but all firms are non-compliant.

While in the bistable regimes only one strategy prevails within each population (i.e. all agents behaves the same way), in the two regimes with an internal stable equilibrium different strategies can coexist in each population. Such an internal stable equilibrium is surrounded by closed trajectories that turn in opposite directions in the two possible regimes: clockwise in one case, counter-clockwise in the other. As explained in the paper, these oscillatory dynamics can be explained using the *prey-predator* conceptual framework. When the trajectories oscillate counterclockwise, honest bureaucrats can be seen as the predators, while non-compliant firms play the role of the preys. On the contrary, when the trajectories oscillate clockwise, dishonest bureaucrats are the predators and compliant firms are the preys. From comparative static analysis of these two dynamic regimes emerges that policy instruments can help the Public Administration reduce both corruption and extortion. However, in an environment characterized by high shares of noncompliant firms and dishonest bureaucrats, an increase in the policy parameters (larger sanctions and higher effort to discover corruption/extortion crimes) has mixed effects on crime deterrence: in the regime with trajectories that turn counter-clockwise, a more stringent policy (higher policy parameters) increases the share of honest bureaucrats, but decreases that of compliant firms, whereas if trajectories turn clockwise, the opposite occurs.

Although the present model is rather simple, in our opinion it manages to provide some interesting and novel insights on the possible dynamics and outcomes that may emerge from the interaction between firms that should comply with the environmental laws and bureaucrats who should verify that this is the case. Moreover, the model provides a general framework that can be easily extended in several directions. To provide an example, further research should be devoted to account for the possibility that the compliant firm may denounce the dishonest bureaucrat who claims a bribe. This possibility would certainly enrich the analysis but it would also further complicate the complex dynamics characterizing the model, therefore it is left for future research.

## Appendix

## **Proof of Proposition 1**

1) In Case (a) (see (5)),  $\Pi_C(h) - \Pi_{NC}(h)$  is strictly increasing in h (see Fig. 1(a)) and the following sub-cases can occur:

i) If:

$$\Pi_C(0) - \Pi_{NC}(0) = b_c - b_e + \theta \ (\eta + s_2) - C_C \ge 0$$

that is, if:

$$s_2 \ge \bar{s}_2 := \frac{C_c + b_e - b_c}{\theta} - \eta \tag{30}$$

then  $\Pi_C(h) - \Pi_{NC}(h) > 0$  holds for every  $h \in (0,1)$  and, consequently, the strategy C dominates the strategy NC.

ii) If:

$$\Pi_C(1) - \Pi_{NC}(1) = ps_1 - C_C \le 0$$

that is, if:

$$s_1 \le \frac{C_C}{p} \tag{31}$$

then  $\Pi_C(h) - \Pi_{NC}(h) < 0$  holds for every  $h \in (0,1)$  and, consequently, the strategy NC dominates the strategy C.

iii) If neither condition (30) nor (31) hold, that is if:

$$s_2 < \bar{s}_2 \quad and \quad s_1 > \frac{C_C}{p} \tag{32}$$

then no strategy dominates the other one, and the graph of the payoff differential  $\Pi_C(h) - \Pi_{NC}(h)$  intersects the *h*-axis at  $h = \bar{h} \in (0, 1)$  (see (4)): for  $h > \bar{h}$  (respectively,  $h < \bar{h}$ ), it holds  $\Pi_C(h) - \Pi_{NC}(h) > 0$  (respectively,  $\Pi_C(h) - \Pi_{NC}(h) < 0$ ).

- 2) In Case (b) (see (6)),  $\Pi_C(h) \Pi_{NC}(h)$  is strictly decreasing in h (see Fig. 1(b)) and the following sub-cases can be observed:
  - i) If:

$$\Pi_C(0) - \Pi_{NC}(0) = b_c - b_e + \theta \ (\eta + s_2) - C_C \le 0$$

that is, if:

 $s_2 \le \bar{s}_2 \tag{33}$ 

then  $\Pi_C(h) - \Pi_{NC}(h) < 0$  holds for every  $h \in (0, 1)$  and, therefore, the strategy NC dominates the strategy C.

ii) If:

$$\Pi_C(1) - \Pi_{NC}(1) = ps_1 - C_C \ge 0$$

that is, if:

$$s_1 \ge \frac{C_C}{p} \tag{34}$$

then  $\Pi_C(h) - \Pi_{NC}(h) > 0$  holds for every  $h \in (0, 1)$  and, therefore, the strategy C dominates the strategy NC.

iii) If neither condition (33) nor condition (34) hold, that is, if:

$$s_2 > \bar{s}_2 \quad and \quad s_1 < \frac{C_C}{p} \tag{35}$$

then no strategy dominates the other one, and the graph of the payoff differential  $\Pi_C(h) - \Pi_{NC}(h)$  intersects the *h*-axis at  $h = \bar{h} \in (0, 1)$  (see (4)): for  $h > \bar{h}$  (respectively,  $h < \bar{h}$ ), it holds  $\Pi_C(h) - \Pi_{NC}(h) < 0$  (respectively,  $\Pi_C(h) - \Pi_{NC}(h) > 0$ ).

## **Proof of Proposition 2**

1) In Case (c) (see (8)),  $\Pi_H(c) - \Pi_D(c)$  is strictly increasing (see Fig. 1(a)) and the following sub-cases can occur:

i) If:

$$\Pi_H(0) - \Pi_D(0) = \theta \sigma_2 - b_c \ge 0$$

that is, if:

$$\sigma_2 \ge \frac{b_c}{\theta} \tag{36}$$

then  $\Pi_H(c) - \Pi_D(c) > 0$  holds for every  $c \in (0, 1)$  and, consequently, the strategy H dominates the strategy D.

ii) If:

$$\Pi_H(1) - \Pi_D(1) = -b_e + \theta \sigma_1 \le 0$$

that is, if:

$$\sigma_1 \le \frac{b_e}{\theta} \tag{37}$$

then  $\Pi_H(c) - \Pi_D(c) < 0$  holds for every  $c \in (0, 1)$  and, consequently, the strategy D dominates the strategy H.

iii) If neither condition (36) nor condition (37) hold, that is if:

$$\sigma_1 > \frac{b_e}{\theta} \quad and \quad \sigma_2 < \frac{b_c}{\theta}$$

$$(38)$$

then no strategy dominates the other one, and the graph of the payoff differential  $\Pi_H(c) - \Pi_D(c)$  intersects the *c*-axis at  $c = \bar{c} \in (0, 1)$  (see (7)): for  $c > \bar{c}$  (respectively,  $c < \bar{c}$ ), it holds  $\Pi_H(c) - \Pi_D(c) > 0$  (respectively,  $\Pi_H(c) - \Pi_D(c) < 0$ ).

2) In Case (d) (see (9)),  $\Pi_H(c) - \Pi_D(c)$  is strictly decreasing (see Fig. 1(b)) and the following sub-cases can be observed:

i) If:

$$\Pi_H(0) - \Pi_D(0) = \theta \sigma_2 - b_c \le 0$$

that is, if:

$$\sigma_2 \le \frac{b_c}{\theta} \tag{39}$$

then  $\Pi_H(c) - \Pi_D(c) < 0$  holds for every  $c \in (0,1)$  and, therefore, the strategy D dominates the strategy H.

$$\Pi_H(1) - \Pi_D(1) = -b_e + \theta \sigma_1 \ge 0$$

that is, if:

$$\sigma_1 \ge \frac{b_e}{\theta} \tag{40}$$

then  $\Pi_H(c) - \Pi_D(c) > 0$  holds for every  $c \in (0,1)$  and, therefore, the strategy H dominates the strategy D.

iii) If neither condition (39) nor condition (40) hold, that is, if:

$$\sigma_1 < \frac{b_e}{\theta} \quad and \quad \sigma_2 > \frac{b_c}{\theta}$$

$$\tag{41}$$

then no strategy dominates the other one, and the graph of the payoff differential  $\Pi_H(c) - \Pi_D(c)$  intersects the *c*-axis at  $c = \bar{c} \in (0, 1)$  (see (7)): for  $c > \bar{c}$  (respectively,  $c < \bar{c}$ ), it holds  $\Pi_H(c) - \Pi_D(c) < 0$  (respectively,  $\Pi_H(c) - \Pi_D(c) > 0$ ).

## References

- Acemoglu, D., Johnson, S., Robinson, J. A., 2012. The colonial origins of comparative development: an empirical investigation: reply. American Economic Review 102 (6), 3077–3110.
- Acemoglu, D., Verdier, T., 2000. The choice between market failure and corruption. American Economic Review 90 (1), 194–211.
- Aidt, T. S., 2009. Corruption, institutions, and economic development. Oxford Review of Economic Policy 25 (2), 271–291.
- Andreozzi, L., 2002. Oscillations in the enforcement of law: an evolutionary analysis. Homo Oeconomicus 18 (3-4), 403–428.
- Andreozzi, L., 2004. Rewarding policemen increases crime. Another surprising result from the inspection game. Public Choice 121 (1-2), 69–82.
- Antoci, A., Borghesi, S., Russu, P., 2005. Biodiversity and economic growth: trade-offs between stabilization of the ecological system and preservation of natural dynamics. Ecological Modelling 189 (3), 333–346.
- Antoci, A., Fiori Maccioni, A., Russu, P., 2016. The ecology of defensive medicine and malpractice litigation. PLoS ONE 11 (3), e0150523.
- Antoci, A., Sacco, P. L., 1995. A public contracting evolutionary game with corruption. Journal of Economics 61 (2), 89–122.
- Antoci, A., Sacco, P. L., 2002. The emergence of deontological codes in public administration. Journal of Mathematical Sociology 26 (4), 271–307.
- Bardhan, P., 1997. Corruption and development: a review of issues. Journal of Economic Literature, 1320–1346.
- Becker, G., 1968. Crime and punishment: an economic approach. Journal of Political Economy 76, 169–217.

- Blackburn, K., Bose, N., Haque, M. E., 2006. The incidence and persistence of corruption in economic development. Journal of Economic Dynamics and Control 30 (12), 2447–2467.
- Blanco, E., Lozano, J., 2015. Ecolabels, uncertified abatement, and the sustainability of natural resources: an evolutionary approach. Journal of Evolutionary Economics 25 (3), 623–647.
- Borghesi, S., 2001. The environmental Kuznets curve: a critical survey. In: Franzini M. and Nicita A. (eds.), Economic Institutions and Environmental Policy. Ashgate. London, UK, pp. 201–224.
- Brander, J. A., Taylor, M. S., 1998. The simple economics of Easter island: a Ricardo-Malthus model of renewable resource use. American Economic Review 88 (1), 119–138.
- Cole, M. A., 2007. Corruption, income and the environment: an empirical analysis. Ecological Economics 62 (3), 637–647.
- Cole, M. A., Elliott, R. J., Fredriksson, P. G., 2006. Endogenous pollution havens: does FDI influence environmental regulations?\*. The Scandinavian Journal of Economics 108 (1), 157– 178.
- Cressman, R., Morrison, W. G., Wen, J.-F., 1998. On the evolutionary dynamics of crime. Canadian Journal of Economics , 1101–1117.
- Damania, R., Fredriksson, P. G., List, J. A., 2003. Trade liberalization, corruption, and environmental policy formation: theory and evidence. Journal of Environmental Economics and Management 46 (3), 490–512.
- D'Amato, A., Mazzanti, M., Nicolli, F., 2015. Waste and organized crime in regional environments: how waste tariffs and the mafia affect waste management and disposal. Resource and Energy Economics 41 (C), 185–201.
- D'Amato, A., Zoli, M., 2012. Illegal waste disposal in the time of the mafia: a tale of enforcement and social well being. Journal of Environmental Planning and Management 55 (5), 637–655.
- D'Amato, A., Zoli, M., 2013. A note on illegal waste disposal, corruption and enforcement. In: D'Amato A., Mazzanti M., Montini A., Waste Management in Spatial Environments. Routledge. Abingdon, UK.
- Demsetz, H., 1967. Toward a theory of property rights. American Economic Review, 347–359.
- Di Gioacchino, D., Franzini, M., 2008. Bureaucrats' corruption and competition in public administration. European Journal of Law and Economics 26 (3), 291–306.
- Di Vita, G., 2007. A note on exogenous changes in incentives for and deterrence of corruption. European Journal of Law and Economics 24 (1), 15–27.
- Di Vita, G., 2011. Curbing corruption in Public Administration: a case study from Italy. International Journal of Public Administration 34 (10), 631–645.
- Di Vita, G., 2014. Environmental crime. In: Backhaus J. (ed.), Encyclopedia of Law and Economics. Springer. New York, USA, pp. 1–8.
- Dinda, S., 2004. Environmental Kuznets curve hypothesis: a survey. Ecological Economics 49 (4), 431–455.

- Dong, F., Wang, P., Wen, Y., 2016. Credit search and credit cycles. Economic Theory 61 (2), 215–239.
- Dreher, A., Herzfeld, T., 2008. The economic costs of corruption: a survey of the empirical evidence. In: Luca F.N. (ed.), Economic corruption: detection, costs and prevention. Chapter 4, Nova Science.
- Fredriksson, P. G., List, J. A., Millimet, D. L., 2003. Bureaucratic corruption, environmental policy and inbound US FDI: theory and evidence. Journal of Public Economics 87 (7), 1407– 1430.
- Fredriksson, P. G., Svensson, J., 2003. Political instability, corruption and policy formation: the case of environmental policy. Journal of Public Economics 87 (7), 1383–1405.
- Friehe, T., 2008. Correlated payoffs in the inspection game: some theory and an application to corruption. Public Choice 137 (1-2), 127–143.
- Garoupa, N., 1997. The theory of optimal law enforcement. Journal of Economic Surveys 11 (3), 267–295.
- Glaeser, E. L., La Porta, R., Lopez-de Silanes, F., Shleifer, A., 2004. Do institutions cause growth? Journal of Economic Growth 9 (3), 271–303.
- Grossman, G. M., Krueger, A. B., 1991. Environmental impacts of a North American free trade agreement. National Bureau of Economic Research, USA.
- Han, L., Strange, W., 2015. The microstructure of housing markets: search, bargaining, and brokerage. In: Duranton G., Henderson V., Strange W. (eds.), Handbook of regional and urban economics. Elsevier, pp. 813–886.
- Hofbauer, J., Sigmund, K., 1988. The theory of evolution and dynamical systems: mathematical aspects of selection. Cambridge University Press.
- Holler, M. J., 1993. Fighting pollution when decisions are strategic. Public Choice 76 (4), 347–356.
- Huntington, S. P., 1968. Political order in changing societies. New Haven: Yale University Press
- Kijima, M., Nishide, K., Ohyama, A., 2010. Economic models for the environmental Kuznets curve: a survey. Journal of Economic Dynamics and Control 34 (7), 1187–1201.
- Kuznets, S., 1955. Economic growth and income inequality. American Economic Review 45 (1), 1–28.
- Leff, N. H., 1964. Economic development through bureaucratic corruption. American Behavioral Scientist 8 (3), 8–14.
- Leitao, A., 2010. Corruption and the environmental Kuznets curve: empirical evidence for sulfur. Ecological Economics 69 (11), 2191–2201.
- Lipset, S. M., 1960. Political man: the social basis of modern politics. NY: Doubleday .
- Lopez, R., Mitra, S., 2000. Corruption, pollution, and the Kuznets environment curve. Journal of Environmental Economics and Management 40 (2), 137–150.

- Lui, F. T., 1985. An equilibrium queuing model of bribery. Journal of Political Economy 93 (4), 760–781.
- Mauro, P., 1998. Corruption and the composition of government expenditure. Journal of Public Economics 69 (2), 263–279.
- Méndez, F., Sepúlveda, F., 2006. Corruption, growth and political regimes: cross country evidence. European Journal of Political Economy 22 (1), 82–98.
- Méon, P.-G., Weill, L., 2010. Is corruption an efficient grease? World Development 38 (3), 244–259.
- Mo, P. H., 2001. Corruption and economic growth. Journal of Comparative Economics 29 (1), 66–79.
- Panayotou, T., 1993. Empirical tests and policy analysis of environmental degradation at different stages of economic development. International Labour Organization. Geneva, Switzerland.
- Pellegrini, L., Gerlagh, R., 2004. Corruption's effect on growth and its transmission channels. Kyklos 57 (3), 429–456.
- Polinsky, A. M., Shavell, S., 2000. The fairness of sanctions: some implications for optimal enforcement policy. American Law and Economics Review 2 (2), 223–237.
- Rogerson, R., Shimer, R., 2011. Search in macroeconomic models of the labor market. In: Card D. Ashenfelter O. (eds.), Handbook of Labor Economics. Vol. 4. North-Holland Amsterdam, pp. 619–700.
- Rosser, J. B., 2011. Complex evolutionary dynamics in urban-regional and ecologic-economic systems: from catastrophe to chaos and beyond. Springer. New York, USA.
- Shafik, N., Bandyopadhyay, S., 1992. Economic growth and environmental quality: time series and cross section evidence. World Development Report. World Bank. Washington DC, USA.
- Svensson, J., 2005. Eight questions about corruption. Journal of Economic Perspectives 19 (3), 19–42.
- Treisman, D., 2000. The causes of corruption: a cross-national study. Journal of Public Economics 76 (3), 399–457.
- Tsebelis, G., 1989. The abuse of probability in political analysis: the Robinson Crusoe fallacy. American Political Science Review 83 (01), 77–91.
- Tsebelis, G., 1990. Are sanctions effective? A game-theoretic analysis. Journal of Conflict Resolution 34 (1), 3–28.
- Weibull, J. W., 1995. Evolutionary Game Theory. MIT Press.
- Welsch, H., 2004. Corruption, growth, and the environment: a cross-country analysis. Environment and Development Economics 9 (05), 663–693.

## Ultimi Contributi di Ricerca CRENoS

I Paper sono disponibili in: http://www.crenos.it

- 16/14 Vittorio Pelligra, Tommaso Reggiani, Daniel John Zizzo, "Responding to (un)reasonable requests"
- 16/13 Pasqualina Arca Gianfranco Atzeni, Luca Deidda, "Asset exemption in entrepreneurs' bankruptcy and the informative role of collateral"
- 16/12 Miguel Casares, Luca Deidda, Jose E. Galdon-Sanchez, "Loan production and monetary policy"
- 16/11 Manuela Pulina, Valentina Santoni, "An analysis on the Italian agricultural firms: effects of public subsidies"
- 16/10 Tiziana Medda, Vittorio Pelligra, Tommaso Reggiani, "Does Experience Affect Fairness, Reciprocity and Cooperation in Lab Experiments?"
- 16/09 Bianca Biagi, Maria Giovanna Brandano, Manuela Pulina, "The effect of tourism taxation: a synthetic control approach"
- 16/08 Edoardo Otranto, Massimo Mucciardi, "A Flexible Specification of Space-Time AutoRegressive Models"
- 16/07 Leonardo Becchetti, Vittorio Pelligra, Francesco Salustri, "Testing for Heterogeneity of Preferences in Randomized Experiments: A Satisfaction-Based Approach Applied to Multiplayer Prisoners' Dilemmas"
- 16/06 Anna Bottasso, Maurizio Conti, Giovanni Sulis, "Firm Dynamics and Employment Protection: Evidence from Sectoral Data "
- 16/05 Oliviero A. Carboni, Giuseppe Medda, "R&D, Export, and Investment Decision"
- 16/04 Leonardo Becchetti, Germana Corrado, Vittorio Pelligra, Fiammetta Rossetti, "Happiness and Preferences in a Legality Social Dilemma Comparing the Direct and Indirect Approach"
- 16/03 Leonardo Becchetti, Vittorio Pelligra, Francesco Salustri, Alejandra Vásquez "Gender differences in Socially Responsible Consumption. An Experimental Investigation"
- 16/02 Leonardo Becchetti, Vittorio Pelligra, Alejandra Vásquez, "Are Women Naturaliter More Cooperative? An Experimental Investigation of the Vote-with-the-Wallet Game"
- **16/01** Leonardo Becchetti, Vittorio Pelligra, Fiammetta Rossetti, "The Corporate Legality Game A Lab Experiment on The Impact of Policies, Frames and Information"
- 15/22 Fabio Cerina, Luisanna Cocco, Katiuscia Mannaro, Michele Marchesi, Francesco Pigliaru, "Insularity and the development of a local network: a simulation model applied to the Italian railway system"
- 15/21 Fabio Cerina, "Is insularity a locational disadvantage? Insights from the New Economic Geography"
- 15/20 Giorgio Garau, Giovanni Mandras, "Economy-wide rebound effects from an increase in efficiency in the use of energy: the Italian case"
- 15/19 Bianca Biagi, Maria Gabriela Ladu, Marta Meleddu, Vicente Royuela, "Tourism and quality of life: a capability approach"
- 15/18 Gianfranco Atzeni, Francesco Chessa, Luca G. Deidda, Malika Hamadi, Stefano Usai, "Access to microcredit and borrowers' behavior: Evidence from Sardinia"

www.crenos.it